Author's response to reviews

Title: Effectiveness of a training program on compliance with recommendations for venous lines care

Authors:

  MARIA JESUS PEREZ-GRANDA (massus@hotmail.es)
  MARIA GUEMBE (mariaguembe@hotmail.com)
  CRISTINA RINCON (crisrsanz@yahoo.es)
  PATRICIA MUÑOZ (pmunoz@hggm.es)
  EMILIO BOUZA (emilio.bouza@gmail.com)

Version: 8 Date: 31 May 2015

Author's response to reviews: see over
Dear Dr. Zingg,

Thank you very much for your comments and suggestions with regard to our article entitled “Effectiveness of a training program in compliance with recommendations for venous lines care”.

1. We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Reply: The language used in the text has been improved as suggested.

2. Please update your ethics statement to include the name of the ethics committee that approved your study.

Reply: The ethics statement has been updated as requested.

3. Clarify if the nurses gave consent to take part.

Reply: This information has been clarified in the text.

Thank you again for your suggestions.

Maria Jesus Pérez